

## G OPEN ACCESS

**Citation:** Lee S, Joo H (2025) Passenger and freight travel patterns: A cluster analysis based on urban networks. PLoS ONE 20(3): e0318084. <u>https://doi.org/10.1371/journal.</u> pone.0318084

**Editor:** Feier Chen, Shanghai Jiao Tong University, CHINA

Received: July 16, 2024

Accepted: January 3, 2025

Published: March 11, 2025

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pone.0318084

**Copyright:** © 2025 Lee, Joo. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data availability statement:** The minimal data set required to replicate our study's

RESEARCH ARTICLE

# Passenger and freight travel patterns: A cluster analysis based on urban networks

## Soyeong Lee<sub>(b</sub>, Heesun Joo(b\*

Department of Urban Engineering, Gyeongsang National University, Jinju, Republic of Korea

\* hsjoo@gnu.ac.kr

## Abstract

While research on population travel patterns and urban networks has been active, it has primarily focused on passenger travel, leaving freight travel relatively underexplored. This study addresses this gap by analyzing both passenger and freight travel patterns, network structures, and central areas. It uses origin-destination (OD) data, considering total travel volume by purpose and mode. The study applies regular equivalence and power centrality to examine differences in human and logistics flows across South Korea from an urban network theory perspective. The key findings are as follows. First, passenger travel, predominantly short-distance, exhibits lower density and intensity than freight travel. Freight travel, on the other hand, demonstrates strong density across short, medium, and long distances, with more travel routes concentrated around nodal regions. Second, passenger travel forms several polynucleated clusters, including short-distance movements. Conversely, freight travel forms a few extensive clusters that encompass medium and longdistance movements. Third, the spatial interaction of passenger travel is influenced by the OD distance, unlike freight travel. Interestingly, the distance between central areas of freight travel can be longer than that of passenger travel. This may stem from the strategic positioning of certain suburban areas as central areas to optimize logistics efficiency. This study emphasizes the importance of morphological and functional linkages between cities by identifying inter-regional differences in passenger and freight flows. It also proposes spatial planning strategies based on urban hierarchy.

## Introduction

Cities grow and maintain their functions not in isolation but through interactions with neighboring cities. Recognizing the importance of these interactions and relationships, there has been a surge in efforts to analyze the role of nodal regions within networks [1-3]. Network analyses have been conducted that utilize various indicators and examine factors contributing to external effects within urban networks [4-9]. However, most city networks exhibit more morphological than functional trends, highlighting the need for an integrative approach [10]. In South Korea, urban decline and population issues are recognized as significant concerns, prompting research into regional patterns of population migration. Addressing these social problems necessitates a systematic approach that takes into account urban functions and structures [11]. This approach is closely tied to urban network formation, indicating a shift in

findings has been uploaded to Figshare and can be accessed at the following DOI: 10.6084/m9.figshare.26953639. The original data used in this study were provided by the Korea Transport Database (KTDB) and can be accessed at <u>https://www.ktdb.go.kr/</u>.

**Funding:** This work was financially supported by the National Research Foundation of Korea (NRF) (<u>https://www.nrf.re.kr</u>.) via the Ministry of Science and ICT (MSIT) (<u>https://www. msit.go.kr</u>) in the form of a grant (RS-2024-00339015) received by HJ. No additional external funding was received for this study. The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

urban form from "place-centered" to "flow-centered" [1,12]. However, most prior studies on population movement have focused on passenger travel [13–19], with freight travel receiving less attention [20,21], largely due to the lack of reliable data [22]. This study aims to elucidate the patterns, network structures, and regional differences in passenger and freight travel, focusing on their impacts on various aspects of a region's economy, society, and culture. In particular, this study focuses on the technical analysis of the static patterns of passenger and freight travel to elucidate the structural characteristics of urban networks in South Korea. The study primarily employs statistical and graphical methods to analyze passenger and freight travel patterns, while excluding any dynamic analyses related to population decline or regional changes.

The specific objectives of this study are as follows:

- 1. Examine the current status and movement patterns of passenger and freight travel with the aim of identifying trends in both human and material flows.
- 2. Identify common sub-regions within passenger and freight travel networks through regular equivalence analysis.
- 3. Identify primary and secondary central areas and propose strategies for enhancing interregional linkages based on the analysis results.

## **Related work**

## Urban network theory

The urban network theory, which has evolved through various concepts such as "dispersed city," "network city," "city network," "polynucleated metropolitan region," and "polycentric or polynuclear urban region," has been linked to the discourse on sustainable urban policy, particularly in the context of the European Union, since the mid-2000s [1,23-27]. This theory posits that cities emerge from the interactions and mutual development of economic, social, and physical fields. A city is deemed to have the potential for self-sustaining development when the cumulative benefits outweigh the disadvantages over time. An urban network system is defined as a collaborative arrangement where two or more independent cities supplement urban functions through transport and communication infrastructure to achieve economies of scale [27].

Recently, the functional linkage networks between large cities and small to medium-sized cities have garnered increased attention. Such networks encompass three core concepts. First, the principle of mutual cooperation suggests that intercity relationships extend beyond hierarchical structures, fostering economic development through city-to-city collaboration [28]. Second, the concept of network externalities indicates that participation in the network can yield economies of scale and synergy effects. Finally, the network is not predicated on the region's hierarchical structure but emerges from intercity connections sharing similar or dissimilar characteristics through long-distance relations between comparably-sized cities.

Cities offer diverse functions through specialization, with high-level functions potentially provided in lower-tier cities under certain circumstances [28,29]. These network cities, also known as polycentric urban regions, are characterized by the coexistence of multiple specialized cities within a specific area. Polycentric urban areas, similar to network cities, are defined by the presence of polynucleated cities within a certain area and the connectivity among them [30]. However, while two or more cities may coexist within a specific area and exhibit inter-dependence regarding certain indicators, they may not necessarily share the area's identity or culture. Furthermore, the expansion of network theory has prompted various studies focusing

on the classification of intercity connections, examination of factors causing externalities in network cities, and investigation of the principles governing the organization of intercity networks [4,7,10].

#### Prior research on the comparison of passenger travel and freight travel

Spatial interaction, one form of which is intercity movement patterns, is consistently defined as the movement and mobility of objects, ideas, goods, and people between spatially separated parts or places. The movement of people and goods within cities forms a dominant pattern of intercity interactions [31]. Numerous discussions have been held on the relationship between travel flow and urban form since the mid-1980s [32-36]. However, most studies have focused on passenger travel, with less attention given to the interaction between various urban area characteristics and freight travel activities. In conclusion, urban networks are constructed not only by human flows but also by material flows that underpin national or regional industries. Therefore, understanding urban networks solely through daily passenger travel is insufficient [37].

Passenger and freight movements are fundamentally different. The former, defined as passenger flow, involves the movement of individuals for various reasons, while the latter, known as freight flow, is solely concerned with transporting goods from one point in the supply chain to another [38]. Analysis of population movement data is typically used to understand the morphological connections between cities, while freight travel data is analyzed to identify functional linkages [7]. As depicted in Fig 1, passenger movement is generally associated with short distances, constrained by the distance/time ratio. In contrast, freight movement encompasses a wider area, closely aligned with the principle of comparative advantage [39]. This distinction underscores the clear difference in characteristics between human mobility and logistics flows [18,40–43].

Intercity passenger and freight flows do not always align [<u>39,43,44</u>]. While passengers may undertake multiple journeys within an urban area daily, amounting to thousands annually, goods are primarily transported from outside the city to the urban area. Following this, they are typically moved to one or two additional locations within the region before



Fig 1. Passenger and freight mobility. Source: Rodrigue JP. The geography of transport systems. 5th ed. Routledge; 2020.

https://doi.org/10.1371/journal.pone.0318084.g001

consumption. Consequently, the spatial interactions of freight movement are minimally affected by the OD distance. The dominant position and network structure within a region can vary, depending on the unique characteristics of each city [37,45]. It has been observed that when travel pattern-related flows are used as a measure of functional connectivity, they form a hierarchical structure [46]. In summary, while most passenger travel tends to gravitate towards nearby destinations [47-49], freight movement is more closely associated with specific suburban areas.

#### Research on network structures and clusters

Research on interregional travel patterns has long been a focus in urban planning [50]. However, previous conclusions were deemed limited due to the lack of large-scale data and appropriate methodologies [51]. The advent of network theory and sensor technology has addressed these limitations, enabling the use of extensive data in various studies to analyze urban systems. These studies have examined diverse areas such as firms and global cities [52–56], finance [19,57,58], transportation, including taxis, railways, and aviation [59–63], cellular base stations and call logs [64–68], population movement and household surveys [69–71], and public transport smart cards [72–76].

Studies employing traditional network indicators can be categorized into three groups: those examining network structure characteristics, spatial structural changes, and network efficiency. Regarding network structure characteristics, social network analysis has been used to study the centrality and urban system of global cities, based on the location information of the headquarters and branches of the world's top 500 multinational corporations [56]. Similarly, Thiemann et al. [57] analyzed the dominance of each city within the global urban system using air passenger data. Liu et al [60] used taxi passenger data in Shanghai to analyze intra-city movement patterns and urban structure, identifying the city's sub-regional structure and characteristics. Saberi et al. [70] compared the network characteristics of passenger demand patterns in Chicago and Melbourne, finding similar characteristics despite differences in topography and urban structure. Lee et al. [64] highlighted the importance of establishing regional living zones by examining movement patterns and structural characteristics of the working population using mobile base station data.

In the spatial structural change context, Alderson et al. [55] analyzed the reconstitution and impact of the global urban system using multinational corporation location data, but failed to provide evidence of increased inequalities between cities. A study analyzing global air passenger flows and cross-border relations suggested that the influence between cities worldwide is somewhat integrated due to the growth of major cities in developing countries and emerging economies [54]. Zhong et al. [75] used smart card data to measure mobility diversity and volatility, finding that this volatility appears at various levels. Guimerà et al. [62] analyzed the structure of the global air transport network, revealing that the structure of various communities within the network, in addition to geographical factors, is critical to predicting centrality. Iqbal et al. [68] analyzed highway travel networks to reveal the correlation between regional economic development and travel patterns, observing changes in spatial interactions between centrality and GDP.

Regarding network efficiency, Salisbury and Barnett [58] identified the centrality of the interregional financial network using fund transfer data between credit cards and banks. Huang et al. [59] reviewed the effect of urban network externalities on urban growth empirically, using train operation frequency data, and determined which city could receive more benefits. Boyd et al. [52] critically analyzed the study by Neal [61], which proposed a new measure of recursive power and recursive centrality in the global urban network, concluding that the existing measure of eigenvector centrality is more useful. Wall and van der Knaap [53]

analyzed the network of the top 100 multinational corporations and subsidiaries in 2,259 cities worldwide, demonstrating how cities are interconnected in various industries and services, and how these connections operate in the economic system.

Urban network theory provides insights into the connections between cities by region, offering theoretical and policy alternatives to address developmental imbalances at both national and regional levels. It serves as an analytical system that can examine how cities are interconnected and how these connections influence various societal aspects, including the economy. However, the theory of network cities has been criticized for its lack of consideration for the real local community economy [77]. It requires systematization of analytical techniques to empirically examine the connectivity of network cities, expansion of connectivity indicators, reconfirmation of the normative nature inherent in network cities, the establishment of cooperative governance for network cities. Even so, the emphasis has shifted toward the functional connectivity of cities rather than their geographical proximity, yielding many studies that apply the concept of network cities and offer theoretical justifications for the effects of network cities as well as empirical research conducted in various aspects.

A review of the literature reveals that most studies have explored complex social phenomena related to urban planning, which is traditionally difficult to access, by actively using large-scale data and network analysis methodologies. These studies recognize the importance of network structures and focus on revealing how intercity connectivity and structural characteristics affect social and economic dynamics, such as regional economic development, travel patterns, and population movement. Studies focusing on network structure characteristics emphasize the analysis of individual cities or structural locations, such as urban centrality, dominance, and sub-regional structure. Studies related to spatial structural changes of networks primarily address structural changes and redistribution of influence that occurs within urban systems over time. Studies on network efficiency differ in that they analyze the flow and efficiency of information, resources, and logistics within the network and focus on empirically reviewing how the network can be optimized and which places are occupied by certain cities within the network.

This study distinguishes itself from others in several ways. First, regarding scope, it compares the morphological and functional network structures using passenger and freight travel data of a broad area, including 17 si/do (cities/provinces) and 250 si/gun/gu (cities/counties/ districts) in South Korea. Second, for multifaceted data analysis, this study uses total travel volume, encompassing various purposes of passenger and freight travel and data by tonnage in its analysis, focusing on specific travel purposes or modes. Third, regarding methodology, regional clustering was performed using REGGE algorithms, and intercity connectivity, hierarchy, and centrality were quantitatively analyzed using power centrality indicators. In summary, this study stands out from previous ones as it comprehensively analyzes the connectivity, regular equivalence, interregional central area, and role between nodal regions of the urban network using a wide spatial scope and passenger and freight travel data of the entire South Korea.

## Data and methods

## Data

This study encompasses all of South Korea, analyzing 17 si/do and 250 si/gun/gu. The 2019 data includes the most recent OD data for passenger and freight travel. The Korea Transport Database (<u>https://www.ktdb.go.kr/</u>) provided the data, which, for passenger travel, comprised OD data on total travel volume, including seven travel purposes such as commuting,

schooling, working, shopping, home returning, leisure/entertainment/visiting relatives, and others. This data was derived from the total travel volume of linked trips between si/gun/gu. For freight travel, the data included OD data of total travel volume, covering light, medium, and heavy tonnage classes, using the total travel volume of unlinked truck trips between si/gun/gu.

The study's objective is to compare the patterns, distance, hierarchical structure, and centrality of interregional human and material networks, necessitating interregional flow data. Therefore, this study utilized the travel OD of 17 si/do and 250 si/gun/gu in South Korea, based on the passenger and freight OD travel volume in 2019. The value was processed as 0 when i = j, using Netminer for analysis. Passenger and freight flows are crucial in defining power centrality. Analyzing travel volume relative to population at nodal points reveals that centrality may change with alterations in administrative districts. Furthermore, differences in trip departure and arrival volume by node mean that the ratio of arrival to departure varies at each point, affecting the centrality of the relevant point. However, the actual effect on centrality can be accurately reflected by calculating and applying the average of total trip departure and arrival volume. To adjust these differences, travel volume was corrected by dividing it by the average population of each region. In the analysis of passenger and freight travel, the number of passengers and vehicles is more significant than the presence of travel, so it was converted to a symmetric matrix before the analysis. Since the number of passengers and vehicles is small compared to the population, the number of passengers per 1,000 was used as the flow rate, as shown in Equation (1).

$$R_{ij} = R_{ji} = \frac{\left(X_{ij} + X_{ji}\right)}{2} / \frac{\left(P_i + P_j\right)}{2} \times 1,000.$$
(1)

This study used threshold settings to construct an OD matrix for analyzing interregional passenger and freight travel. An alternative method involves calculating the ratio of trip inflow (outflow) for each origin (destination) to the total trip inflow (outflow) for each destination (origin) and extracting only the connection lines where the ratio is 5% or more [78]. However, this method has a drawback: the travel volume, which accounts for 5%, varies greatly depending on the size of the total trip inflow (outflow) at each nodal point. Therefore, this study established the number of connection lines by setting an absolute threshold after excluding the internal travel volume of each nodal point. The thresholds were set at 1,000 trips for passenger travel and 100 trips for freight travel. If the travel volume between two nodal points is less than 1,000 trips or 100 trips, the value is set to '0'; for values above the threshold, a phase structure was built using Equation (1). This method allows for the construction of a phase structure that accurately reflects the actual flow between nodal points where at least a certain level of interaction occurs.

**Methods.** The research methodology is divided into three sections. The first involves identifying travel patterns to ascertain the strength of intercity flows by measuring the connectivity between nodal regions within passenger and freight networks. The second step involves identifying the regular equivalence of urban networks to analyze the similarity of positions or characteristics among 17 si/do and group them accordingly. This step utilizes REGGE algorithms to measure the similarity between nodal regions. The third step involves deriving primary and secondary central areas to highlight the significance and roles of regions, as defined by the proportion, function, and interregional relationships determined by the interregional passenger and freight travel network patterns. This step measures the importance of nodal regions within the network using power centrality indices for each clustered region. Chapter 2 of this study explores urban network theory and reviews various

practical studies related to network and cluster analysis. Chapter 3 outlines the methodologies for empirical analysis, while Chapter 4 presents the results of cluster and centrality analysis, including the current status. Chapter 5 provides implications (see Fig 2).

**Regular equivalence (REGGE).** Location analysis in social network analysis aims to identify actors connected in similar patterns within the network (i.e., individuals or regions exhibiting similar relationship forms) and classify them into comparable groups [79]. Consequently, regions in the same location play similar roles or hold similar positions. In the context of travel networks, a region's position reflects its connectivity within the country. The concept of regular equivalence is applied here to measure the network position of regions [80,81]. This concept mirrors the structure of relationships between different regions based on the similarity of the positions and roles performed by regions within the network, making regular equivalence a crucial measure for understanding intercity roles and positions in network analysis [82].

Levels of regular equivalence are measured using REGGE algorithms [82,83]. These algorithms estimate the similarities between two specific regions when analyzing interregional relationships. The process begins with an initial estimate of the similarities in interregional travel. Then, estimates of similarity between the two regions are adjusted by examining how similar the two regions are to other regions to which they are each connected. By repeating this process multiple times, quantitative figures on how similar and interchangeable the two regions are to each other are obtained. The scale of regular equivalence generated by REGGE algorithms is measured by Equation (2) [82–84].



Fig 2. Research flow.

https://doi.org/10.1371/journal.pone.0318084.g002

$$M_{ij}^{t+1} = \frac{\sum_{k=1}^{g} \max_{m=1}^{g} \sum_{r=1}^{R} (i_{jr} M_{kmr}^{t} + j_{jr} M_{kmr}^{t})}{\sum_{k=1}^{g} \max_{m} * \sum_{r=1}^{R} (i_{jr} Max_{kmr} + j_{jr} Max_{kmr})},$$
(2)

where  $M_{ii}^{t+1}$  denotes the regular equivalence between regions i and j at the (t+1)th iteration, based on the travel network. The denominator signifies the maximum possible consistency when the connections of region i to all other regions (k) perfectly align with the connections of region j to all other regions (m). Here, the changing entities for regions i and j (k and m) exhibit regular similarity. The numerator optimally aligns the relationship of region j with m to that of region i with k, a process weighted by the regular equivalence of k and m from the previous iteration. Consequently, this algorithm identifies the optimal match in the connections between region i and all other regions, and between region j and all other regions. These connections are weighted according to their similarity with other regions in the network and are calculated by the maximum theoretically possible consistency [85]. Thus,  $M_{ii}^{t+1}$ , the regular equivalence, represents the inverse function of the degree to which the connections of region i with other regions align with the connections of region j with other regions. The similarity of interregional travel flows is reassessed after each iteration [85], and the degree of similarity between the regions in the network is quantified. The resulting similarity ranges from 0 to 1, where 0 signifies complete nonequivalence between two regions, and 1 indicates perfect regular equivalence.

## Network centrality analysis (power centrality)

Centrality in a network signifies an individual's influence and power within that network. Analyzing centrality allows for the identification of key actors within the network, or those receiving significant attention. These influential actors are often referred to as central or hub nodes. The primary centrality indices in a network include degree centrality, closeness centrality, and betweenness centrality [86]. Other indices, such as eigenvector centrality [87], page rank [88], and structural holes [89], also exist. Regardless of the index used, centrality is not an absolute measure; instead, it indicates relative ranking. Therefore, normalized centrality values may be employed, depending on the network's scale (e.g., size and density). Power centrality, a new centrality index, was developed to address the variability in results produced by different centrality indices [90]. Also known as Bonacich beta centrality, it supplements degree and eigenvector centrality. In this context,  $\beta$  (beta) represents the weight needed to calculate Bonacich's centrality index. A small  $\beta$  value emphasizes the local structure surrounding specific actors, while a large  $\beta$  value considers the entire network structure.

The centrality of a region can be analyzed by treating the origin and destination as nodes in the network, representing interactions between nodes as links, and examining the movement flow within the network [91,92]. High centrality in adjacent regions suggests the potential activation of economic and cultural activities in the region. Interregional interactions are often confrontational, leading to a negative  $\beta$  value. However, previous research set  $\beta$  as 0 (ignoring the impact of indirect connections requiring multiple steps in degree centrality) or  $1/(\lambda \max)$  (maximizing the impact of indirect connections requiring multiple steps in eigenvector centrality, where  $\lambda$  is the maximum eigenvalue). For travel networks with confrontational interactions, these centrality indices may sometimes yield inconsistent results. Therefore, power centrality can be calculated as shown in Equation (3).

$$\mathbf{c}_{i}(\alpha,\beta) = \sum_{j} (\alpha + \beta \mathbf{c}_{j}) \mathbf{R}_{ij}, \qquad (3)$$

where  $c_i$  is a vector containing the beta centrality of nodes as elements,  $\alpha$  is the range used to standardize centrality values,  $\beta$  represents the range of weights based on the distance with nodes, and  $R_{ij}$  is an element of the adjacency matrix. The  $\beta$  value, as described above, assigns more weight to the network structure surrounding a specific high-centrality point when calculating power centrality. This suggests that larger  $\beta$  values place more emphasis on a broader network structure. Consequently, the  $\beta$  value cannot be uniformly determined when nodal points represent individual regions. By setting  $\beta = -1/(\lambda \max)$ , assuming confrontational interregional interaction, it is possible to relatively reduce the impact of the distant network structure compared to  $\beta = 1/(\lambda \max)$ , as shown in Equation (4).

$$C_{\beta} = \alpha \sum_{k=0}^{\infty} \beta^{k} R^{k+1} = \alpha \Big( R1 - \beta R^{2} 1 + \beta^{2} R^{3} 1 - \beta^{3} R^{4} 1 \cdots \Big).$$
(4)

## Results

## Travel status

As depicted in <u>Table 1</u> and <u>Fig 3</u>, South Korea's passenger travel in 2019 comprised 13,849,020 inter-si/do trips and 45,963,003 inter-si/gun/gu trips daily. The capital regions (Seoul Special City, Incheon Metropolitan City, Gyeonggi-do) accounted for at least 50% of the total travel volume, with most travel concentrated in this region. Incheon Metropolitan City and Gyeonggi-do have more arrival volume than departure volume, while Seoul Special City has more departure volume than arrival volume. This suggests that Seoul, a hub for business, education, and cultural activities, draws people from its satellite cities, Incheon and Gyeonggi-do. The high trip departure volume in Seoul indicates that many individuals residing in Incheon or Gyeonggi-do commute to Seoul for work or school. When analyzed by si/gun/gu,

	( <b>I</b> )								
Classification	Movements bet	ween si/do		Movements bet	Movements between si/gun/gu				
	Departure volume		Arrival volume	Arrival volume		Departure volume		Arrival volume	
Seoul	3,846,725	28%	3,760,801	27%	13,029,681	28%	12,943,757	28%	
Busan	547,385	4%	539,030	4%	4,026,211	9%	4,017,857	9%	
Daegu	507,775	4%	508,036	4%	2,832,127	6%	2,832,388	6%	
Incheon	1,046,503	8%	1,119,696	8%	2,663,049	6%	2,736,242	6%	
Gwangju	301,996	2%	296,925	2%	1,728,014	4%	1,722,943	4%	
Daejeon	406,773	3%	384,253	3%	1,624,349	4%	1,601,829	3%	
Ulsan	219,291	2%	202,596	1%	1,206,679	3%	1,189,985	3%	
Gyeonggi	3,994,031	29%	4,019,733	29%	9,931,947	22%	9,957,649	22%	
Gangwon	207,832	2%	210,812	2%	623,109	1%	626,089	1%	
Chungbuk	350,926	3%	341,639	2%	1,210,940	3%	1,201,653	3%	
Chungnam	496,964	4%	492,487	4%	1,359,585	3%	1,355,108	3%	
Jeonbuk	168,907	1%	167,538	1%	1,068,468	2%	1,067,100	2%	
Jeonnam	328,891	2%	332,072	2%	924,370	2%	927,552	2%	
Gyeongbuk	600,712	4%	611,529	4%	1,365,236	3%	1,376,053	3%	
Gyeongnam	577,136	4%	594,407	4%	1,923,938	4%	1,941,208	4%	
Jeju	40,119	0%	40,792	0%	238,243	1%	238,915	1%	
Sejong	207,055	1%	226,674	2%	207,055	0%	226,674	0%	
Total	13,849,020	100%	13,849,020	100%	45,963,003	100%	45,963,003	100%	

Table 1. Passenger travel status (Trip/day, %)

https://doi.org/10.1371/journal.pone.0318084.t001



Note: A method exists for representing at least 5% of travel when forming connection lines in the travel network. However, the travel volume, which constitutes this 5%, can greatly vary depending on the total trip inflow (outflow) at each nodal point. Therefore, this study constructed a flow map by excluding the internal travel volume at each nodal point and setting an absolute threshold of 1,000 trips.

#### Fig 3. Passenger travel flows and patterns.

https://doi.org/10.1371/journal.pone.0318084.g003

the travel patterns were similar to those at the si/do level, but with a greater concentration in metropolitan cities (Busan Metropolitan City, Daegu Metropolitan City) and the Seoul Capital Area. Gyeongsangnam-do, despite being adjacent to three metropolitan cities (Busan, Daegu, and Ulsan), had higher travel volume only in certain regions near these cities.

Freight travel, as shown in <u>Table 2</u> and <u>Fig 4</u>, was 1,052,474 inter-si/do vehicles and 2,259,505 inter-si/gun/gu vehicles daily in 2019. The capital regions accounted for at least 45%

Classification	Movements be	tween si/do		Movements be	Movements between si/gun/gu				
	Departure volume		Arrival volum	Arrival volume		Departure volume		Arrival volume	
Seoul	126,366	12%	123,670	12%	318,418	14%	315,722	14%	
Busan	67,538	6%	68,427	7%	159,074	7%	159,962	7%	
Daegu	30,707	3%	32,733	3%	94,447	4%	96,472	4%	
Incheon	98,779	9%	95,486	9%	186,170	8%	182,877	8%	
Gwangju	20,912	2%	21,360	2%	49,940	2%	50,388	2%	
Daejeon	19,189	2%	19,150	2%	55,747	2%	55,709	2%	
Ulsan	25,944	2%	27,576	3%	51,764	2%	53,397	2%	
Gyeonggi	249,235	24%	262,007	25%	597,146	26%	609,918	27%	
Gangwon	40,807	4%	37,768	4%	70,408	3%	67,369	3%	
Chungbuk	59,075	6%	60,152	6%	91,758	4%	92,836	4%	
Chungnam	75,000	7%	73,439	7%	118,480	5%	116,919	5%	
Jeonbuk	38,678	4%	36,005	3%	68,445	3%	65,771	3%	
Jeonnam	45,031	4%	43,186	4%	89,492	4%	87,647	4%	
Gyeongbuk	64,930	6%	62,722	6%	120,625	5%	118,418	5%	
Gyeongnam	80,238	8%	79,058	8%	163,543	7%	162,364	7%	
Jeju	0	0%	0	0%	14,004	1%	14,004	1%	
Sejong	10,045	1%	9,733	1%	10,045	0%	9,733	0%	
Total	1,052,474	100%	1,052,474	100%	2,259,505	100%	2,259,505	100%	

#### Table 2. Freight travel status (Vehicle/day, %).

https://doi.org/10.1371/journal.pone.0318084.t002

of the total freight travel volume. Gyeonggi-do receives more freight travel than it sends out, suggesting that it serves as a consumption point for manufacturing businesses, logistics, and distribution centers, or as a hub for storage and distribution. This implies that goods produced in Seoul and Incheon are likely imported by firms in Gyeonggi-do or redistributed to other regions through Gyeonggi-do. In contrast, Seoul Special City and Incheon Metropolitan City are production, export, or departure points as they have more freight travel departures than arrivals. This is likely due to import and export activities through Incheon's international airport and port, as well as various commercial and industrial activities in Seoul. The travel volume gap between the Seoul Capital Area and non-capital areas such as Gyeongsangnam-do, Chungcheongnam-do, Busan Metropolitan City, Gyeongsangbuk-do, and Chungcheongbuk-do was narrower for freight than for passenger travel, revealing different patterns in human and logistics networks. Finally, the si/gun/gu analysis yielded similar results to the si/do level. The flow map of si/gun/gu showed a more even exchange of influence with surrounding areas than passenger travel, indicating a functional trade-off.

In summary, travel volume significantly differs across regions. Seoul, Daejeon, Chungnam, and Jeonbuk, characterized by higher departure volume than arrival volume in both passenger and freight travel, serve as economic and social hubs. These regions are home to diverse industries, including technology, manufacturing, and agriculture, leading to the active production and distribution of goods and services. The high departure volume of freight in these regions suggests a vibrant logistics sector. Conversely, Daegu and Gyeonggi, where arrivals surpass departures in both passenger and freight travel, function as centers of consumption and distribution. Therefore, while Seoul, Daejeon, Chungnam, and Jeonbuk are primary producers of goods and services, Daegu and Gyeonggi are major consumers, illustrating the complementary economic roles of these regions.

Regions with high passenger travel but low freight departures, such as logistics, manufacturing, or agriculture, may be hubs of human resources, indicating fewer industries that



Note: The process for establishing connection lines for the passenger travel network mirrors the aforementioned method. This study created a flow map by excluding the internal travel volume at each nodal point and setting an absolute threshold of 100 trips.

Fig 4. Freight travel flows and patterns.

https://doi.org/10.1371/journal.pone.0318084.g004

generate significant freight compared to other regions. This highlights the diversity of each region's economic role and industrial structure. Regions like Incheon, Gangwon, Jeonnam, Gyeongbuk, Gyeongnam, Jeju, and Sejong, except for Incheon, generally have a low population density and are known for manufacturing, agriculture, fishing, and specialized industries. These regions have higher departure volume than the arrival volume for freight travel,

suggesting active trade or distribution, with goods or resources produced within the region being moved to other regions.

Incheon, home to a major international trade port, plays a crucial role in collecting and redistributing goods domestically and internationally. In contrast, Busan, Gwangju, Ulsan, and Chungbuk, characterized by high population density and economic activities centered on commerce, services, and light industries, may attribute their high passenger travel departure volume to these activities. A comparison of passenger and freight travel patterns reveals that passenger travel patterns have a lighter intercity travel density than freight travel patterns and exhibit balanced and even patterns, resembling a distributed connected set. The extensive interactions in these regions may be attributed to the human network's close relation to travel routes, such as subways, express bus terminals, train stations, and airports, in addition to cars. Freight travel patterns, however, display greater density and intensity than passenger travel patterns, with more long-distance movements observed due to the clear purpose of goods movement.

This interpretation closely aligns with the composition of South Korea's gross domestic product (GDP). Approximately 60% of South Korea's GDP is generated by the services sector, while around 30% is derived from manufacturing. This economic structure can result in variations in passenger and freight transportation between cities, depending on their economic activities and industrial structure. For instance, a city like Seoul, which serves as the administrative hub for major corporations, is likely to experience an increase in passenger travel related to the service sector. In contrast, a city like Incheon, which hosts a major international port, is likely to experience more active freight transport due to its role as a hub for imports and exports.

## Travel network equivalence analysis

Network analysis offers intriguing insights into the sub-structures within a network, serving as a structural foundation for stratification. Passenger and freight travel networks, for instance, comprise numerous interconnected subnetworks or travel blocks. These blocks are more substantially linked to regions within the same network than to those in other network blocks. Travel network analysis reveals core regions with tight connections and peripheral regions with looser connections. This study aims to explore the sub-structures of regions in each block by categorizing the core and peripheral regions between the origin and destination within the travel network. The X and Y coordinates represent the eigenvalues derived from the clustering matrix, with regions within the shortest average network distance included in a single cluster. The regular equivalence analysis of the passenger travel network (Fig 5) reveals five clusters from eight groups for inter-si/do movements. These clusters are located in Quadrant 4 (Cluster 1) with high trip departure and arrival volume, Quadrant 2 (Cluster 1) with low trip departure and arrival volume, Quadrant 3 (Cluster 3) with high trip departure volume but low arrival volume, and Quadrant 1 (Clusters 4 and 5) with high trip arrival volume but low departure volume. For movements between si/gun/gu, six clusters were formed from eight groups, located in Quadrant 4 (Clusters 2, 3, and 6) with high trip departure and arrival volume, Quadrant 3 (Cluster 1) with high trip departure volume but low arrival volume, and Quadrant 1 (Clusters 4 and 5) with high trip arrival volume but low departure volume.

The regular equivalence analysis of the freight travel network (Fig 6) shows five clusters from six groups for inter-si/do movements. These clusters are located in Quadrant 4 (Clusters 1 and 3) with high trip departure and arrival volume, Quadrant 2 (Cluster 5) with low trip departure and arrival volume, Quadrant 3 (Cluster 2) with high trip departure volume but low arrival volume, and Quadrant 1 (Cluster 4) with high trip arrival volume but low departure volume. For movements between si/gun/gu, four clusters were formed from six

Classification		Move	ements be	etween si	do	Movements between si/gun/gu						
	Der	ndrogram	Level         8         0.031 (social           Number         17         16         15           Dargely         Groupset(1)         -         -           Brand(2)         Groupset(1)         -         -           Brand(2)         Groupset(1)         -         -           Changbad(10)         -         Changbad(10)         -           Changbad(10)         -         -         -           App(10)         -         -         -           Jointon(1)         -         -         -		₩ • 99 • 97 • 199 • 129 •	59 UP 239 99 179	Lord 2 and 2					
Cluster		MDS	4.91 - 4.91 - 4.95 - 4.	on over oza Joobb Joobb Joenam	0.07 0.35 0.43	0.01 1.19 1.47 Lacheon Gyrenggi Scoul	4.67 4.12 4.35 4.3 4.45 4.45 4.45 4.45 4.45 4.45	25 25 25 Jack	स 6.12 ०.14 ve	013 0.11 0.4 Changuan Ganguon Sejon Dag Jeju	con Joenbuk gum Joenbuk Gyoengenk Gyoengenk San	
	Output Summary		MEAN	STD. DEV	MIN	MAX	Output	MEAN	STD. DEV	MIN	MAX	
			0.29	0.19	0.02	0.84	Summary	0.15	0.13	0.01	0.74	
	C1	GG, SU	0.32	0.34	0.00	0.94	GG, SU	0.67	0.22	0.05	0.94	
	C2	GJ, JN	0.38	0.35	0.00	0.99	GJ, JN	0.33	0.21	0.04	0.90	
Descriptive	C3	GN, GB, DG, BS	0.18	0.21	0.01	0.98	GN, BS, US	0.37	0.29	0.00	0.94	
Statistics	C4	GW, CN, CB	0.32	0.34	0.00	0.94	GW, SJ	0.39	0.19	0.01	0.79	
	C5	DJ, SJ, JB	0.28	0.34	0.00	0.95	DJ, JB, CN, CB	0.16	0.17	0.00	0.92	
	C6	US		-	-	-	GB, DG	0.28	0.25	0.01	0.92	
	C7	IC	-	-	-	-	IC	-	-	-	-	
	C8	JJ	-	-	-	-	JJ	-	-	-	-	

\* BS: Busan, CB: Chungbuk, CN: Chungnam, DG: Daegu, DJ: Daejeon, GB: Gyeongbuk, GG: Gyeonggi, GJ: Gwangju, GN: Gyeongnam, GW: Gangwon, IC: Incheon, JB: Jeonbuk, JJ: Jeju, JN: Jeonnam, SJ: Sejong, SU: Seoul, US: Ulsan

#### Fig 5. Passenger REGGE cluster dendrogram and multidimensional scaling results.

https://doi.org/10.1371/journal.pone.0318084.g005

groups, located in Quadrant 4 (Clusters 3 and 4) with high trip departure and arrival volume, Quadrant 3 (Cluster 1) with high trip departure volume but low arrival volume, and Quadrant 1 (Cluster 2) with high trip arrival volume but low departure volume. The passenger travel network analysis results show clear quadrant clustering, grouping morphologically similar regions. However, the freight travel network analysis results differ, grouping similar regions even across different quadrants, indicating functional connections.

## Centrality analysis results

Power centrality was calculated to measure regional centrality, employing interregional passenger and freight flows (Equation 1). The initial results, presented in Fig 7, identify the primary and secondary central areas for both passenger and freight travel among si/do. Chungnam and Sejong emerged as the regions where both types of travel shared the same primary and secondary central areas. Gwangju and Gyeongnam, while also sharing central areas, differed in their order of precedence. In some regions, only one central area coincided with both types of travel. For instance, Daejeon and Jeonbuk shared the same primary central area for both passenger and freight travel. Similarly, Gyeongbuk, Chungbuk,



\* BS: Busan, CB: Chungbuk, CN: Chungnam, DG: Daegu, DJ: Daejeon, GB: Gyeongbuk, GG: Gyeonggi, GJ: Gwangju, GN: Gyeongnam, GW: Gangwon, IC: Incheon, JB: Jeonbuk, JJ: Jeju, JN: Jeonnam, SJ: Sejong, SU: Seoul, US: Ulsan

#### Fig 6. Freight REGGE cluster dendrogram and multidimensional scaling results.

https://doi.org/10.1371/journal.pone.0318084.g006

and Gangwon had the same primary central area for passenger travel and secondary for freight travel.

Moreover, Gyeonggi and Daegu shared the secondary central area for passenger travel and the primary for freight travel. However, no regions were observed where the secondary central areas for both types of travel coincided. Despite these similarities, Seoul, Jeonnam, and Busan exhibited different central areas, and no comparison targets were available for Incheon and Ulsan. Specifically, the differences in the influence and ranking of primary and secondary central areas in passenger and freight travel networks across various cities and provinces can be analyzed. For passenger traffic, the primary central area with the highest influence is Gyeongsan-si in Gyeongbuk (C3), scoring 4.610, followed by Suseong-gu in Daegu (C3) with a score of 3.357, Gangnam-gu in Seoul (C2) with 3.221, Bundang-gu in Seongnam-si, Gyeonggi (C2) with 3.093, Naju-si in Jeonnam (C2) with 2.196, and Buk-gu in Gwangju (C2) with 2.033. For freight travel, Bucheon-si in Gyeonggi (C1) tops the list with a score of 4.354, followed by Gyeongsan-si in Gyeongbuk (C2) with 3.716, Bupyeong-gu in Incheon (C1) with 3.530, Gangseo-gu in Busan (C3) with 3.312, Dong-gu in Daejeon (C2) with 3.164, and Gimhae-si in Gyeongnam (C3) with 3.115. Interestingly, while Gyeongsan-si in Gyeongbuk

Classification			Passeng	er travel			Freigh	t travel				
Class	sification	First central	areas	Second centra	l areas	Frist central	areas	Second central	areas			
	1	Seoul Gangnamgu	3.221	Seoul Seochogu	2.123	Gyeonggi Bucheonsi	4.354	Gyeonggi Gimposi	2.000			
C1	2	Gyeonggi Seongnamsi Bundanggu	3.093	Gyeonggi Bucheonsi	2.107	Incheon Bupyeonggu	3.530	Incheon Seogu	2.727			
	3	-	-		-	Seoul Gangseogu	2.224	Seoul Gurogu	1.097			
	1	Jeonnam Najusi	2.196	Jeonnam Mokposi	1.558	Gyeongbuk Gyeongsansi	3.716	Gyeongbuk Yeongcheonsi	1.558			
	2	Gwangju Bukgu	2.033	Gwangju Gwangsangu	1.396	Daegu Donggu	3.164	Daegu Dalseogu	1.964			
C2	3	-		-	-	Gwangju Gwangsangu	2.041	Gwangju Bukgu	0.561			
	4	-	-	-	-	Ulsan Uljugun	1.208	Ulsan Namgu	0.733			
	5	-		-1	-	Jeonnam Gwangyangsi	1.013	Jeonnam Jangseonggun	0.868			
	6	-		ы	8	Jeonbuk Gunsansi	0.304	Jeonbuk Namwonsi	0.293			
	1	Gyeongbuk Gyeongsansi	4.610	Gyeongbuk Chilgokgun	1.169	Busan Gangseogu	3.312	Busan Sasanggu	0.819			
C2	2	Daegu Suseonggu	3.357	Daegu Donggu	2.420	Gyeongnam Gimhaesi	3.115	Gyeongnam Yangsansi	1.701			
CS	3	CyeongnamYa ngsansi	0.676	Gyeongnam Gimhaesi	0.593	-	×	-	-			
	4	Busan Bukgu	0.420	Busan Geumjeonggu	0.385	-	÷	Ξ	-			
	1	Chungbuk Jecheonsi	1.455	Chungbuk Cheongjusi Seowongu	1.351	Gangwon Wonjusi	2.705	Gangwon Yeongwolgun	0.646			
C4	2	Chungnam Cheonansi Seobukgu	1.350	Chungnam Cheonansi Dongnamgu	1.098	Chungbuk Chungjusi	2.615	Chungbuk Jecheonsi	1.057			
	3	Gangwon Yeongwolgun	1.273	Gangwon Goseonggun	1.052	Chungnam Cheonansi Seobukgu	0.226	Chungnam Cheonansi Dongnamgu	0.174			
	1	Sejong Sejongsi	1.855	-	-	Sejong Sejongsi	1.789	-	-			
C5	2	Daejeon Yuseonggu	1.346	Daejeon Seogu	1.128	Daejeon Yuseonggu	1.016	Daejeon Daedeokgu	0.981			
	3	Jeonbuk Gunsansi	0.022	-	-	-	-	-	-			
Comparison				The first central (the primary cen     The scond central (the scond are calculated as the scond central scond are calculated as the scond central scond are calculated as the scond as the scond are calculated as the scond are calculated as the scond are calculated as the scond as the sco	areas ters) ral areas enters)			<ul> <li>The first central (the primary central (the primary central))</li> <li>The second central (the second are central)</li> </ul>	o areas ters) ral areas enters)			

Fig 7. Comparison of power centrality in movements between si/do.

https://doi.org/10.1371/journal.pone.0318084.g007

has the most significant influence in passenger traffic, it ranks second in freight travel, following Gyeonggi. In Daegu, Suseong-gu serves as the central area for passenger traffic, while Dong-gu holds this position for freight travel. Similarly, in Gyeonggi, while Bundang-gu in Seongnam-si is the central area for passenger traffic, Bucheon-si takes this role for freight travel.

The centrality analysis of movements between si/do reveals a concentration of influence in regions C1, C2, and C3. This suggests that the primary and secondary central areas in these regions play a pivotal role in the transport and logistics network. Unexpectedly, power centrality was highly rated in Gyeongbuk's Gyeongsan-si and Jeonnam's Naju-si, which may be attributed to their unique geographical locations. Gyeongsan-si, due to its proximity to Daegu Metropolitan City, benefits from the economic and social advantages of the metropolitan area while maintaining its own industrial and cultural activities. Naju-si, home to the headquarters of the Korea Electric Power Corporation, has developed as a central hub for science, technology, and the energy industry, distinguishing it from other regions in Jeollanam-do. This suggests that high centrality in non-metropolitan regions may not only be a result of traditional transport infrastructure development or logistics system efficiency, but also be related to the concentrated accumulation of industrial and technological innovation, and the enhancement of intra- and inter-regional connectivity. Conversely, regions C4 and C5 exhibited relatively low centrality, indicating less developed connectivity and influence within the transport and logistics network. This could be due to a variety of factors, including inadequate transport infrastructure, inefficient logistics systems, and restrictions on economic activities, which may hinder regional development and population influx. To enhance the connectivity and influence of these regions, policy support is necessary for the expansion of transport infrastructure, the improvement of logistics systems, and the strengthening of interregional networks.

The second set of results, as shown in Fig 8, identifies the first and second central areas of passenger and freight travel clusters in various si/gun/gu regions. Gwangju and Ulsan were the regions where the first and second central areas for both passenger and freight travel were identical. In contrast, Chungnam had the same central areas, but the order was reversed. In cases where only one central area was the same for both types of travel, Daejeon had the same first central areas, while Gyeongbuk had the same first central area for passenger travel and the second for freight travel. Daegu had the same second central area for passenger travel and the first for freight travel. Gyeonggi and Chungbuk had the same second central areas for both types of travel. Despite these similarities, Seoul, Jeonnam, Gyeongnam, Busan, Gangwon, and Jeonbuk had completely different central areas for passenger and freight travel. Incheon had no comparison target as it only had clusters for freight travel.

Specifically, the differences in the influence and ranking of primary and secondary central areas in passenger and freight travel networks across various regions can be analyzed. In passenger travel, the primary central area was Daejeon Seo-gu (C5) with a score of 3.959, followed by Seoul Gangnam-gu (C1) with 3.527, Gwangju Buk-gu (C2) with 2.787, Busan Busanjin-gu (C3) with 2.611, Daegu Suseong-gu (C6) with 2.513, and Gangwon Sokcho-si (C4) with 2.271. In contrast, for freight travel, Gwangju Buk-gu (C2) had the highest score of 3.998, followed by Gyeongnam Gimhae-si (C3) with 3.701, Busan Gangseo-gu (C3) with 3.263, Chungnam Cheonan-si Dongnam-gu (C4) with 2.884, Incheon Namdong-gu (C1) with 2.584, and Gyeonggi Hwaseong-si (C1) with 2.268. Interestingly, Gwangju Buk-gu ranked third in passenger travel but first in freight travel.

In Busan, Busanjin-gu was the primary area for passenger travel, while Gangseo-gu was primary for freight travel. The centrality analysis of movements between regions indicates an even distribution within clusters for both passenger and freight travel. This suggests that the transport and logistics system functions effectively and extensively across the nation, without bias toward specific regions. Particularly, the high influence of Gwangju Buk-gu and Gwangsan-gu, Daejeon Seo-gu and Yuseong-gu, and Busan Busanjin-gu and Gangseo-gu, in both passenger and freight travel, underscores the centrality of transportation and goods movement in these regions. Therefore, it is crucial to enhance the network's efficiency and connectivity with surrounding areas via intensive monitoring of human and material flows.

			Passeng	er travel			Freigh	t travel	
Classi	fication	First central	areas	Second cent	ral areas	Frist central a	reas	Second centr	al areas
	1	Seoul Gangnamgu	3.527	Seoul Seochogu	2.816	Incheon Namdonggu	2.584	Incheon Junggu	2.503
CI	2	Gyeonggi Seongnamsi Bungdanggu	1.547	Gyeonggi Bucheonsi	0.852	Gyeonggi Hwaseongsi	2.268	Gyeonggi Bucheonsi	2.206
	3		-	-	-	Seoul Gangseogu	1.321	Seoul Yangcheongu	1.066
	1	Gwangju Bukgu	2.787	Gwangju Gwangsangu	2.785	Gwangju Bukgu	3.998	Gwangji Gwangsangu	3.915
	2	Jeonnam Najusi	0.441	Jeonnam Mokposi	0.311	Jeonnam Gwangyangsi	1.221	Jeonnam Jangseonggun	0.849
C2	(1)	1-11	-		-	Daejeon Seogu	1.200	Daejeon Daedeokgu	1.065
	4	-	-	-	-	Ulsan Namgu	0.264	Ulsan Junggu	0.259
	5	-		-	-	Jeonbuk Jeoneupsi	0.209	Jeonbuk Iksansi	0.205
	1	Busan Busanjingu	2.611	Busan Haeundaegu	2.143	Gyeongnam Gimhaesi	3.701	Gyeongnam Yangsansi	2.625
	2	Ulsan Namgu	1.824	Ulsan Junggu	1.317	Busan Gangseogu	3.263	Busan Sasanggu	1.596
C3	()	Gyeongnam Changwonsi Seongsangu	0.402	Gyeongnam Changwonsi Uichanggu	0.379	Daegu Dalseogu	2.237	Daegu Dalseonggun	1.763
	(4)	-		-	-	Gyeongbuk Gumisi	0.846	Gyeongbuk Gyeongsansi	0.823
	1	Gangwon Sokchosi	2.271	Gangwon Goseonggun	1.767	Chungnam Cheonansi Dongnamgu	2.884	Chungnam Cheonansi Seobukgu	2.806
C4	2	-	-	-	-	Chungbuk Cheongjusi Cheongwongu	2.215	Chungbuk Cheongjusi Seowongu	2.177
	())	-	-	-	-	Gangwon Wonjusi	0.422	Gangwon Gangneungsi	0.311
	1	Daejeon Seogu	3.959	Daejeon Yuseonggu	3.263	-	-	-	-
	2	Chungnam Cheonansi Seobukgu	0.943	Chungnam Cheonansi Dongnamgu	0.814	-	-	-	-
C5	3	Jeonbuk Jeonjusi Deokjingu	0.802	Jeonbuk Jeonjusi Wansangu	0.774	-		-	×
	4	Chungbuk Cheongusi Heungdeokgu	0.777	Chungbuk Cheongjusi Seowongu	0.736	-	-	-	-
66	1	Daegu Suseonggu	2.513	Daegu Dalseogu	2.406	-	1	147	-
0	2	Gyeongbuk Gyeongsansi	1.071	Gyeongbuk Chilgokgun	0.469	-	5-1	-	
Comparison				The first central disc primary central disc primary central disc primary central disc primary central disc prime second central disc prime seco	areas tters) ral areas			The first central: the primary central: the primary central: the second central:	areas ters) al areas

Fig 8. Comparison of power centrality in movements between si/gun/gu.

https://doi.org/10.1371/journal.pone.0318084.g008

In summary, metropolitan areas with high population density and well-developed transport infrastructure, such as Daejeon Seo-gu and Seoul Gangnam-gu, ranked high for passenger travel. These areas have a high demand for movement and serve as hubs of daily life and economic activities. For freight travel, regions with production and logistics facilities, such as Gwangju Buk-gu and Gyeongnam Gimhae-si, showed high influence. This indicates that freight travel is closely tied to the flow of economic activities, including production, processing, and distribution. However, the primary and secondary central areas for passenger and freight travel were identical in some regions, such as Chungnam and Sejong, but differed in others, such as Seoul, Jeonnam, Gyeongnam, and Busan. Factors such as economic activities, industrial structures, and the respective regions' geographical characteristics may affect these variations.

## Discussion

This study analyzes inter-regional travel flows from an urban network perspective using passenger and freight travel data across South Korea, identifying key differences between passenger and freight travel patterns. The analysis of the origin and destination patterns of passenger and freight travel revealed that passenger transport occurs across a wider range of regions compared to freight travel, with a more pronounced imbalance between the volume of departures and arrivals.

Passenger travel primarily occurs for personal reasons, such as commuting, shopping, and leisure. Travel therefore occurs over a wide spatial range that includes not only densely populated large cities, but also medium-sized cities and surrounding areas. This perspective aligns with the findings presented in the study by Adams [3], which emphasizes that cities function as interconnected components of a larger system rather than existing independently. Understanding the relationships between cities is essential for effective urban planning and development. When the volume of arrivals exceeds that of departures in passenger travel, the region may be strengthening its role as a destination. This travel pattern also aligns with the view presented in Adams [3], which states that such interactions form a broad network. As demonstrated in Fig 9, the observed passenger travel patterns show dispersed and interconnected flows across regions, forming a cohesive and extensive travel network that supports Adams' assertion. This indicates that passenger travel largely relies on short-distance movements, forming dispersed travel routes for various purposes and thereby establishing an extensive travel network. In contrast, freight travel is concentrated around specific regions with active economic activities, such as industrial complexes, logistics centers, and ports, focusing on connections between specific routes and areas based on logistical efficiency and economic purposes. Freight travel reflects the physical logistics flow of moving goods from production sites to consumption areas, and this primarily takes place over long-distance routes. Regions that serve as hubs of economic activity tend to have higher departure volumes as they are actively involved in the connections between production, processing, and consumption. This aligns with Meijer's [27] urban network theory, which argues that inter-city cooperation optimizes logistics flows and functions as an efficient system. As shown in Fig 10, the freight



Passenger Travel Patterns (Morphologically Monocentric + Functionally Polycentric)

#### Fig 9. Passenger travel patterns.

https://doi.org/10.1371/journal.pone.0318084.g009



Freight Travel Patterns (Morphologically Polycentric + Functionally Monocentric)

#### Fig 10. Freight travel patterns.

https://doi.org/10.1371/journal.pone.0318084.g010



Fig 11. Passenger travel network clusters.

https://doi.org/10.1371/journal.pone.0318084.g011

travel patterns exhibit concentrated flows between major logistics hubs, demonstrating how inter-city collaboration enhances efficiency within the network.

It is also worth noting the reversal phenomenon observed between passenger and freight travel. In regions with active passenger travel, freight departure volumes tend to be relatively low, and vice versa. This phenomenon is closely linked to the economic characteristics and industrial structure of the respective regions. For example, densely populated areas primarily function as human resource centers, but if industries such as logistics and manufacturing are not well-developed, the volume of freight departures tends to be low. In contrast, in regions where agriculture or manufacturing is developed, freight travel tends to be more active, even with a relatively smaller population, as goods and resources produced in these areas are transported elsewhere. This aligns with the argument presented by Burger and Meijers [10], which suggests that inter-city networks are functionally complementary and can expand economic scale through cooperation. As highlighted in Fig 12, the complementary roles of passenger and freight travel clusters illustrate how distinct functional contributions enhance the overall economic integration of urban networks.

A comparison of the cluster formation ranges for passenger and freight travel using regular equivalence in network analysis revealed significantly different cluster formation patterns for the two types of travel. Freight travel clusters were formed over a wider geographic range compared to passenger travel clusters, but the number of clusters was





Fig 12. Freight travel network clusters.

https://doi.org/10.1371/journal.pone.0318084.g012

relatively smaller. This is consistent with the findings of Frenken and Hoekman [4], who stated that urban networks focus on connecting specific routes and economic hubs. Our analysis of freight travel confirms this pattern, as major routes connecting industrial complexes, ports, and logistics centers serve as focal points within the network, as illustrated in Fig 12. This characteristic of freight travel is primarily centered around specific nodal areas, such as industrial complexes, ports, and logistics centers where logistics and economic activities are active. This aligns with Meijers' [27] explanation that urban network systems are structured based on cooperation between politically independent cities through transportation and communication infrastructure. In contrast, passenger travel tended to form more clusters within a smaller geographic range, possibly because passenger travel forms complex travel patterns that are driven by various individual purposes to different destinations, such as commuting, attending school, shopping, and leisure. These movements are largely influenced by individual decision-making and people's everyday activity ranges. This aligns with Batten [1], who stated that personal travel patterns in urban systems are dispersed, as well as Burger and Meijers [10], who suggested that passenger travel forms clusters due to connections to various destinations rather than concentrations along specific routes. As shown in Fig 11, passenger travel clusters are distributed over a relatively narrow range, reflecting the diverse purposes and localized nature of these trips. Therefore, while freight travel is concentrated on efficiently connecting specific regions that are closely linked to economic activities, passenger travel reflects more diverse and complex individual mobility patterns, resulting in a differentiated network structure with multiple clusters.

The power centrality analysis revealed a difference in the distance between the primary and secondary central areas for passenger travel and freight travel. The central areas of passenger travel are primarily formed around densely populated and economically active regions that have a relatively short distance between the primary and secondary central areas. This can be interpreted as passenger travel primarily relying on short-distance movements between regions, with relatively dense central areas forming due to the diverse travel purposes, such as commuting, shopping, and leisure. In contrast, freight travel tended to show a greater distance between the primary and secondary central areas. This is consistent with Capello's [28] assertion that freight travel follows long-distance economic flows, as well as González [18], who suggested that central areas are formed based on logistical efficiency and economic objectives. Fig 14 illustrates how the central areas of freight travel are situated in regions with developed

logistical infrastructure, reinforcing Capello's and González's arguments. The central areas for freight travel are located in regions with well-developed logistics and manufacturing activities, such as production sites, industrial complexes, logistics centers, and ports. These distances are shaped by logistical efficiency and economic objectives. Therefore, freight travel places importance on the connectivity between specific routes and economic hubs, which suggests that central areas can be formed over a wider range.

This difference is interpreted as stemming from the distinct purposes and requirements of passenger and freight travel. Passenger travel reflects short-distance movements within people's daily activity ranges, while freight travel aims to create economic value through the



Fig 13. Distance between central areas of the passenger travel network.

https://doi.org/10.1371/journal.pone.0318084.g013





https://doi.org/10.1371/journal.pone.0318084.g014

long-distance distribution of raw materials, manufactured goods, and consumer products. These differences in travel patterns demonstrate that the two modes of transport play distinct roles within the urban network. Fig 13 and Fig 14 show contrasting spatial distributions and centrality measures for passenger and freight travel, respectively, underscoring their distinct contributions to the network's economic efficiency, as also supported by prior studies [24,26,30].

## Conclusion

This study examines inter-regional travel flows in South Korea, focusing on passenger and freight travel patterns, central area formation, and strategies to improve regional connectivity. Through empirical data and network analysis, it identifies distinct characteristics of passenger and freight travel, highlighting their unique roles within the urban network.

First, passenger travel exhibited a higher volume and was characterized by short-distance movements that supported everyday activities such as commuting, shopping, and leisure. As shown in Fig 9, passenger travel formed dispersed and interconnected flows across regions, with densely populated areas often serving as central destinations due to a higher volume of arrivals than departures. In contrast, freight travel was concentrated in economically active regions, such as industrial complexes, logistics centers, and ports, with long-distance movements dominating the network. Fig 10 illustrates how freight travel primarily connects production sites to consumption areas, reflecting its role in optimizing logistics and supporting industrial activity.

Second, structural differences between the networks were evident. Passenger travel formed numerous clusters within smaller geographic ranges, as depicted in Fig 11, indicating diverse and localized mobility patterns. Freight travel, on the other hand, exhibited fewer clusters that spanned broader geographic areas, emphasizing efficient connections between economic hubs, as shown in Fig 12. These distinctions underscore that passenger travel supports localized social and economic interactions, while freight travel focuses on long-distance logistical efficiency.

Third, power centrality analysis revealed contrasting spatial distributions of central areas. Passenger travel central areas were densely distributed in regions with high population density and economic activity, with relatively short distances between primary and secondary centers (Fig 13). Conversely, freight travel demonstrated greater distances between central areas, highlighting its focus on connecting regions of production and consumption over long distances (Fig 14). These findings confirm that passenger travel enhances regional accessibility, while freight travel facilitates inter-regional economic integration.

These results underscore the necessity of developing complementary networks that integrate passenger and freight travel to achieve balanced regional development. Passenger travel networks, characterized by multi-nodal activity centers, should be strategically linked with freight travel hubs to optimize logistical flows and improve regional accessibility. For example, enhancing accessibility between densely populated areas and industrial complexes through integrated public transportation and logistics systems can address network inefficiencies and enhance regional productivity.

Additionally, the findings highlight the importance of planning for morphological and functional polycentricity. Passenger travel clusters, driven by diverse social activities, and freight travel clusters, centered on logistical efficiency, must be strategically coordinated to maximize regional productivity. Targeted investments in infrastructure and policies that strengthen these connections can foster balanced development between cities and regions.

In conclusion, this study provides empirical evidence of the distinct travel patterns and network roles of passenger and freight transport through detailed analysis and visualization. By aligning transportation and logistics strategies with these roles, policymakers can design more integrated and sustainable regional planning frameworks. These findings serve as a critical foundation for evidence-based decision-making to enhance inter-regional interactions and promote balanced development across South Korea.

## Limitations and future research

This study provides foundational data for enhancing regional transport infrastructure and economic activities by distinguishing regions where the central areas of passenger and freight travel do and do not coincide. This distinction allows for an examination of connectivity between densely populated and economically active areas by identifying differing movement trends in passenger and freight travel. A quantitative assessment of these central areas and their influence can inform policy decisions aimed at constructing an efficient urban network and optimizing logistics systems. This study also offers insights into the complex dynamics of passenger and freight travel, which are integral to various social and economic activities, and proposes an analysis for developing an efficient transport and logistics system from an urban network perspective. However, this study, which analyzes interregional movement patterns and central areas of passenger and freight travel, has certain limitations. First, it relies on single-year data from 2019, which, while offering a snapshot in time, does not capture the dynamic trends of changing travel patterns over time.

Second, the analysis method, which uses total travel volume data, provides broad-scale results but lacks depth in understanding detailed aspects such as specific purposes or travel volume in specific regions. Finally, the spatial scope of the study is limited. While the analysis at the level of si/gun/gu reveals broad regional characteristics, it fails to identify microscopic network characteristics at a more detailed level.

To address these limitations, future research should consider the following directions: examining changes in socioeconomic situations and the resulting flows in human and material networks using multi-year data to understand more dynamic travel patterns; conducting more granular analyses, such as passenger travel according to purpose, freight travel based on resource movement, and micro-geographic units; and undertaking additional research on transport connectivity and economic revitalization in areas with relatively low centrality to promote balanced regional development. This will establish a basis for policy recommendations to address regional imbalance and propose efficient strategies for developing shrinking cities.

## Acknowledgments

The authors thank the anonymous reviewers and editors for their valuable and constructive suggestions for improving this article.

## Author contributions

Conceptualization: Soyeong Lee. Data curation: Soyeong Lee. Formal analysis: Soyeong Lee. Funding acquisition: Heesun Joo. Investigation: Soyeong Lee. Methodology: Soyeong Lee. Project administration: Heesun Joo. Resources: Heesun Joo.

Supervision: Heesun Joo.

Validation: Soyeong Lee.

Visualization: Soyeong Lee.

Writing – original draft: Soyeong Lee.

Writing – review & editing: Heesun Joo.

## References

- 1. Batten DF. Network cities: creative urban agglomerations for the 21st Century. Urban Stud. 1995;32(2):313–27. https://doi.org/10.1080/00420989550013103
- Hohenverg P, Lees L. The making of Urban Europe. Cambridge: Harvard University Press; 1985;1000–994.
- Adams P. Network Topologies and Virtual Place. Ann Assoc Am Geograp. 1998;88(1):88–106. <u>https://doi.org/10.1111/1467-8306.00086</u>
- Frenken K, Hoekman J. Convergence in an enlarged europe: the role of network cities. Tijd voor Econ Soc Geog. 2006;97(3):321–6. https://doi.org/10.1111/j.1467-9663.2006.00523.x
- Johansson B, Quigley J. Agglomeration and networks in spatial economies. Berlin Heidelberg: Springer 20041–200.
- 6. Meijers E. Polycentric urban regions and the quest for synergy: is a network of cities more than the sum of the parts?. Urban Stud. 2005;42(4):765–81. https://doi.org/10.1080/00420980500060384
- Limtanakool N, Dijst M, Schwanen T. A theoretical framework and methodology for characterising national urban systems on the basis of flows of people: empirical evidence for france and germany. Urban Stud. 2007;44(11):2123–45. https://doi.org/10.1080/00420980701518990
- Glaeser EL, Ponzetto GAM, Zou Y. Urban networks: Connecting markets, people, and ideas. Papers Reg Sci. 2016;95(1):17–60. https://doi.org/10.1111/pirs.12216
- Suárez M, Delgado J. Is Mexico City Polycentric? A Trip Attraction Capacity Approach. Urban Stud. 2009;46(10):2187–211. https://doi.org/10.1177/0042098009339429
- Burger M, Meijers E. Form follows function? linking morphological and functional Polycentricity. Urban Stud. 2011;49(5):1127–49. <u>https://doi.org/10.1177/0042098011407095</u>
- Gu Y, Kraak M-J, Engelhardt Y, Mocnik F-B. A classification scheme for static origin–destination data visualizations. Int J Geograp Inform Sci. 2023;37(9):1970–97. <u>https://doi.org/10.1080/13658816.2023.</u> 2234001
- Camagni RP, Salone C. Network urban structures in northern italy: elements for a theoretical framework. Urban Stud. 1993;30(6):1053–64. https://doi.org/10.1080/00420989320080941
- Yan X-Y, Wang W-X, Gao Z-Y, Lai Y-C. Universal model of individual and population mobility on diverse spatial scales. Nat Commun. 2017;8(1):1639. <u>https://doi.org/10.1038/s41467-017-01892-8</u> PMID: 29158475
- Ren Y, Ercsey-Ravasz M, Wang P, González MC, Toroczkai Z. Predicting commuter flows in spatial networks using a radiation model based on temporal ranges. Nat Commun. 2014;55347. <u>https://doi.org/10.1038/ncomms6347</u> PMID: 25373437
- Yan X-Y, Han X-P, Wang B-H, Zhou T. Diversity of individual mobility patterns and emergence of aggregated scaling laws. Sci Rep. 2013;32678. <u>https://doi.org/10.1038/srep02678</u> PMID: 24045416
- Simini F, González MC, Maritan A, Barabási A-L. A universal model for mobility and migration patterns. Nature. 2012;484(7392):96–100. https://doi.org/10.1038/nature10856 PMID: 22367540
- Song C, Qu Z, Blumm N, Barabási A-L. Limits of predictability in human mobility. Science. 2010;327(5968):1018–21. https://doi.org/10.1126/science.1177170 PMID: 20167789
- González MC, Hidalgo CA, Barabási A-L. Understanding individual human mobility patterns. Nature. 2008;453(7196):779–82. https://doi.org/10.1038/nature06958 PMID: 18528393
- 19. Brockmann D, Hufnagel L, Geisel T. The scaling laws of human travel. Nature. 2006;439(7075):462–5. https://doi.org/10.1038/nature04292 PMID: 16437114
- Rahimi M, Asef-Vaziri A, Harrison R. An inland port location-allocation model for a regional intermodal goods movement system. Marit Econ Logist. 2008;10(4):362–79. https://doi.org/10.1057/mel.2008.17

- Anas A, Liu Y. A regional economy, land use, and transportation model (relu-tran©): formulation, algorithm design, and testing\*. J Reg Sci. 2007;47(3):415–55. <u>https://doi.org/10.1111/j.1467-9787.2007.00515.x</u>
- Yanhong F, Xiaofa S. Research on freight truck operation characteristics based on GPS Data. Procedia Soc Behav Sci. 2013;96:2320–31. https://doi.org/10.1016/j.sbspro.2013.08.261
- Burton I. A restatement of the dispersed city hypothesis. Ann Assoc Am Geograp. 1963;53(3):285–9. https://doi.org/10.1111/j.1467-8306.1963.tb00449.x
- Camagni R, Stabilini S, Diappi L. City networks in the Lombardy region: an analysis in terms of communication flows. Flux. 1994;10(15):37–50. <u>https://doi.org/10.3406/flux.1994.975</u>
- Dieleman FM, Faludi A. Polynucleated metropolitan regions in Northwest Europe: Theme of the special issue. Eur Plan Stud. 1998;6(4):365–77. https://doi.org/10.1080/09654319808720468
- Kloosterman RC, Musterd S. The polycentric urban region: towards a research agenda. Urban Stud. 2001;38(4):623–33. <u>https://doi.org/10.1080/00420980120035259</u>
- Meijers E. From central place to network model: theory and evidence of a paradigm change. Tijd voor Econ Soc Geog. 2007;98(2):245–59. https://doi.org/10.1111/j.1467-9663.2007.00394.x
- Capello R. The city network paradigm: measuring urban network externalities. Urban Stud. 2000;37(11):1925–45. https://doi.org/10.1080/713707232
- **29.** Sohn J. Network city as a new urban growth model: a review on its formation, spatial structure, management, and growth potential. J Korean Geograp Soc. 2011;46:181–96.
- Bailey N, Turok I. Central scotland as a polycentric urban region: useful planning concept or chimera?. Urban Stud. 2001;38(4):697–715. <u>https://doi.org/10.1080/00420980120035295</u>
- D'Este G. Urban freight movement modeling. Handbooks Trans. 2007633–47. <u>https://doi.org/10.1108/9780857245670-033</u>
- 32. Banister D. Unsustainable transport: city transport in the new century. London: Routledge; 2005.
- Breheny M. Densities and sustainable cities: The UK experience. Cities for the new millennium. London: Routledge; 2014.
- 34. ECOTEC. Reducing transport emissions through land use planning. London: HMSO; 1993.
- Newman PWG, Kenworthy JR. The transport energy trade-off: Fuel-efficient traffic versus fuel-efficient cities. Transp Res Part A General. 1988;22(3):163–74. <u>https://doi.org/10.1016/0191-2607(88)90034-9</u>
- 36. Taylor I, Sloman L. Masterplanning checklist. 2008.
- Lee B, Yim S. An analysis of urban network in Seoul metropolitan area by interaction indices. J KARG. 2014;2030–48.
- Woudsma C. Understanding the movement of goods, not people: issues, evidence and potential. Urban Stud. 2001;38(13):2439–55. https://doi.org/10.1080/00420980120094605
- 39. Rodrigue JP. The geography of transport systems. 5th ed. Routledge 2020.
- Liu Y, Kang C, Gao S, Xiao Y, Tian Y. Understanding intra-urban trip patterns from taxi trajectory data. J Geogr Syst. 2012;14(4):463–83. https://doi.org/10.1007/s10109-012-0166-z
- Noulas A, Scellato S, Lambiotte R, Pontil M, Mascolo C. A tale of many cities: universal patterns in human urban mobility. PLoS One. 2012;7(5):e37027. <u>https://doi.org/10.1371/journal.pone.0037027</u> PMID: 22666339
- Liang X, Zhao J, Dong L, Xu K. Unraveling the origin of exponential law in intra-urban human mobility. Sci Rep. 2013;32983. https://doi.org/10.1038/srep02983 PMID: 24136012
- Zhao K, Musolesi M, Hui P, Rao W, Tarkoma S. Explaining the power-law distribution of human mobility through transportation modality decomposition. Sci Rep. 2015;59136. <u>https://doi.org/10.1038/ srep09136</u> PMID: <u>25779306</u>
- Allen J, Browne M, Cherrett T. Investigating relationships between road freight transport, facility location, logistics management and urban form. J Trans Geography. 2012;2445–57. <u>https://doi.org/10.1016/j.jtrangeo.2012.06.010</u>
- Eom H, Woo M. The Impact of Network with Central City on Urban Growth. jkpa. 2019;54(3):15–26. https://doi.org/10.17208/jkpa.2019.06.54.3.15
- 46. Ogunsanya AA. Spatial pattern of urban freight transport in Lagos metropolis. Transp Res Part A General. 1982;16(4):289–300. <u>https://doi.org/10.1016/0191-2607(82)90056-5</u>
- Liu Y, Sui Z, Kang C, Gao Y. Uncovering patterns of inter-urban trip and spatial interaction from social media check-in data. PLoS One. 2014;9(1):e86026. <u>https://doi.org/10.1371/journal.pone.0086026</u> PMID: 24465849

- Masucci AP, Serras J, Johansson A, Batty M. Gravity versus radiation models: on the importance of scale and heterogeneity in commuting flows. Phys Rev E Stat Nonlin Soft Matter Phys. 2013;88(2):022812. <u>https://doi.org/10.1103/PhysRevE.88.022812</u> PMID: <u>24032888</u>
- 49. Yin J, Soliman A, Yin D, Wang S. Depicting urban boundaries from a mobility network of spatial interactions: a case study of Great Britain with geo-located Twitter data. Int J Geograp Inf Sci. 2017;31(7):1293–313. https://doi.org/10.1080/13658816.2017.1282615
- 50. Handy S. Methodologies for exploring the link between urban form and travel behavior. Transp Res Part D: Transp Environ. 1996;1(2):151–65. https://doi.org/10.1016/s1361-9209(96)00010-7
- Yue Y, Lan T, Yeh AGO, Li Q-Q. Zooming into individuals to understand the collective: a review of trajectory-based travel behaviour studies. Travel Behav Soc. 2014;1(2):69–78. <u>https://doi.org/10.1016/j.</u> tbs.2013.12.002
- Boyd JP, Mahutga MC, Smith DA. Measuring centrality and power recursively in the world city network: a reply to neal. Urban Studies. 2013;50(8):1641–7. <u>https://doi.org/10.1177/0042098012466599</u>
- Wall RS, van der Knaap GA. Sectoral differentiation and network structure within contemporary worldwide corporate networks. Econ Geograp. 2011;87(3):267–308. <u>https://doi.org/10.1111/j.1944-8287.2011.01122.x</u>
- Mahutga MC, Xiulian Ma, Smith DA, Timberlake M. Economic globalisation and the structure of the world city system: the case of airline passenger data. Urban Stud. 2010;47(9):1925–47. <u>https://doi.org/10.1177/0042098010372684</u>
- Alderson AS, Beckfield J, Sprague-Jones J. Intercity relations and globalisation: the evolution of the global urban hierarchy, 1981—2007. Urban Stud. 2010;47(9):1899–923. <u>https://doi.org/10.1177/0042098010372679</u>
- Alderson AS, Beckfield J. Power and position in the world city system. Am J Soc. 2004;109(4):811–51. https://doi.org/10.1086/378930
- 57. Thiemann C, Theis F, Grady D, Brune R, Brockmann D. The structure of borders in a small world. PLoS One. 2010;5(11):e15422. https://doi.org/10.1371/journal.pone.0015422 PMID: 21124970
- Joseph G. T. Salisbury, George A. B. The world system of international monetary flows: a network analysis. Inform Soc. 1999;15(1):31–49. https://doi.org/10.1080/019722499128655
- Huang Y, Hong T, Ma T. Urban network externalities, agglomeration economies and urban economic growth. Cities. 2020;107:102882. https://doi.org/10.1016/j.cities.2020.102882
- Liu X, Gong L, Gong Y, Liu Y. Revealing travel patterns and city structure with taxi trip data. J Transp Geograp. 2015;43:78–90. https://doi.org/10.1016/j.jtrangeo.2015.01.016
- **61.** Neal ZP. From central places to network bases: a transition in the u.s. urban hierarchy, 1900–2000. City Community. 2011;10(1):49–75. https://doi.org/10.1111/j.1540-6040.2010.01340.x
- Guimerà R, Mossa S, Turtschi A, Amaral LAN. The worldwide air transportation network: anomalous centrality, community structure, and cities' global roles. Proc Natl Acad Sci U S A. 2005;102(22):7794–9. https://doi.org/10.1073/pnas.0407994102 PMID: <u>15911778</u>
- Smith D, Timberlake M. Hierarchies of dominance among world cities: a network approach. In: Sassen S, (Editor) Global networks linked cities. New York: Routledge; 2002
- Lee S, Bae M, Joo H. Establishment of commuting areas in the workplace population using network analysis. JAIK. 2022;38:23–34.
- Çolak S, Lima A, González MC. Understanding congested travel in urban areas. Nat Commun. 2016;7:10793. https://doi.org/10.1038/ncomms10793 PMID: 26978719
- Alexander L, Jiang S, Murga M, González MC. Origin–destination trips by purpose and time of day inferred from mobile phone data. Transp Res Part C Emerg Technol. 2015;58:240–50. <u>https://doi.org/10.1016/j.trc.2015.02.018</u>
- Toole JL, Colak S, Sturt B, Alexander LP, Evsukoff A, González MC. The path most traveled: Travel demand estimation using big data resources. Transp Res Part C Emerg Technol. 2015;58:162–77. https://doi.org/10.1016/j.trc.2015.04.022
- Iqbal MdS, Choudhury CF, Wang P, González MC. Development of origin–destination matrices using mobile phone call data. Transp Res Part C Emerg Technol. 2014;40:63–74. <u>https://doi.org/10.1016/j.</u> trc.2014.01.002
- 69. Li B, Gao S, Liang Y, Kang Y, Prestby T, Gao Y, et al. Estimation of regional economic development indicator from transportation network analytics. Sci Rep. 2020;10(1):2647. <u>https://doi.org/10.1038/</u> s41598-020-59505-2 PMID: 32060351

- Saberi M, Mahmassani HS, Brockmann D, Hosseini A. A complex network perspective for characterizing urban travel demand patterns: graph theoretical analysis of large-scale origin–destination demand networks. Transportation. 2016;44(6):1383–402. <u>https://doi.org/10.1007/s11116-016-9706-6</u>
- Vasanen A. Functional polycentricity: examining metropolitan spatial structure through the connectivity of urban Sub-centres. Urban Stud. 2012;49(16):3627–44. <u>https://doi.org/10.1177/0042098012447000</u>
- Alsger A, Tavassoli A, Mesbah M, Ferreira L, Hickman M. Public transport trip purpose inference using smart card fare data. Transp Res Part C Emerg Technol. 2018;87:123–37. <u>https://doi.org/10.1016/j.trc.2017.12.016</u>
- Zhao Z, Koutsopoulos HN, Zhao J. Individual mobility prediction using transit smart card data. Transp Res Part CEmerg Technol. 2018;89:19–34. https://doi.org/10.1016/j.trc.2018.01.022
- Yildirimoglu M, Kim J. Identification of communities in urban mobility networks using multilayer graphs of network traffic. Transp Res Procedia. 2017;27:1034–41. <u>https://doi.org/10.1016/j.</u> trpro.2017.12.070
- Zhong C, Manley E, Müller Arisona S, Batty M, Schmitt G. Measuring variability of mobility patterns from multiday smart-card data. J Comput Sci. 2015;9:125–30. <u>https://doi.org/10.1016/j.jocs.2015.04.021</u>
- 76. Sun L, Axhausen KW, Lee D-H, Huang X. Understanding metropolitan patterns of daily encounters. Proc Natl Acad Sci U S A. 2013;110(34):13774–9. <u>https://doi.org/10.1073/pnas.1306440110</u> PMID: 23918373
- 77. Kloosterman RC, Lambregts B. Clustering of economic activities in polycentric urban regions: the case of the randstad. Urban Studies. 2001;38(4):717–32. https://doi.org/10.1080/00420980120035303
- 78. Berry B, Horton F. Geographic perspectives on urban systems. Englewood Cliffs: Prentice-Hall 1970.
- Borgatti SP, Everett MG. Notions of position in social network analysis. Soc Method. 1992;22:1. <u>https://doi.org/10.2307/270991</u>
- White DR, Reitz KP. Graph and semigroup homomorphisms on networks of relations. Soc Netw. 1983;5(2):193–234. https://doi.org/10.1016/0378-8733(83)90025-4
- Borgatti SP, Everett MG. The class of all regular equivalences: Algebraic structure and computation. Soc Netw. 1989;11(1):65–88. <u>https://doi.org/10.1016/0378-8733(89)90018-x</u>
- Wasserman S, Faust K. Social network analysis: methods and applications. Cambridge: Cambridge University Press; 1994.
- **83.** White DR, Reitz KP. Measuring role distance: structural, regular and relational equivalence. Unpublished manuscript. Irvine: University of California; 1985.
- Žiberna A. Direct and indirect approaches to blockmodeling of valued networks in terms of regular equivalence. J Mathe Soc. 2008;32(1):57–84. https://doi.org/10.1080/00222500701790207
- Mahutga MC. The persistence of structural inequality? a network analysis of international trade, 1965-2000. Soc Forces. 2006;84(4):1863–89. https://doi.org/10.1353/sof.2006.0098
- Freeman LC. Centrality in social networks conceptual clarification. Soc Netw. 1978;1(3):215–39. https://doi.org/10.1016/0378-8733(78)90021-7
- Bonacich P. Factoring and weighting approaches to status scores and clique identification. J Mathe Soc. 1972;2(1):113–20. https://doi.org/10.1080/0022250x.1972.9989806
- Brin S, Page L. The anatomy of a large-scale hypertextual Web search engine. Comput Netw ISDN Systems. 1998;30(1–7):107–17. <u>https://doi.org/10.1016/s0169-7552(98)00110-x</u>
- Burt R. Structural holes: The social structure of competition. Cambridge: Harvard University Press 1992.
- 90. Bonacich P. Power and centrality: a family of measures. Am J Soc. 1987;92(5):1170–82. <u>https://doi.org/10.1086/228631</u>
- Borgatti SP. Centrality and network flow. Soc Netw. 2005;27(1):55–71. <u>https://doi.org/10.1016/j.socnet.2004.11.008</u>
- 92. Friedkin NE. Theoretical Foundations for Centrality Measures. Am J Soc. 1991;96(6):1478–504. https://doi.org/10.1086/229694